Author's response to reviews

Title: Would school closure for the 2009 H1N1 influenza epidemic have been worth the cost?: a computational simulation of Pennsylvania

Authors:

Shawn T Brown (stb60@pitt.edu)
Julie H Y Tai (juliehy@pitt.edu)
Rachel R Bailey (rrb16@pitt.edu)
Philip C Cooley (pcc@rti.org)
William D Wheaton (wdw@rti.org)
Margaret A Potter (mapotter@pitt.edu)
Ronald E Voorhees (rvoorhees@achd.net)
Megan LeJeune (lejeune.megan@gmail.com)
John J Grefenstette (gref@pitt.edu)
Donald S Burke (donburke@pitt.edu)
Sarah M McGlone (smm168@pitt.edu)
Bruce Y Lee (byl1@pitt.edu)

Version: 4  Date: 11 May 2011

Author's response to reviews:

Addressed all of the concerns from the editorial committee:

1. Made sure the abstract was updated.
2. Fixed the broken link to the CDC H1N1 planning document.
3. Checked document again for all spelling and grammar.

Additionally:
1. Modified the Acknowledgments section with appropriate language for the CDC grant.